Author’s response to reviews

Title: Neonatal diet impacts liver mitochondrial bioenergetics in piglets fed formula or human milk

Authors:

Eugenia Carvalho (ECarvalho@uams.edu)
Sean Adams (shadams@uams.edu)
Elisabeth Borsheim (EBorsheim@uams.edu)
Michael Blackburn (BlackburnMichaelL@uams.edu)
Kikumi Ono-Moore (OnomooreKD@archildrens.org)
Matthew Cotter (cotterm@archildrens.org)
Anne Bowlin (Abowlin@uams.edu)
V.Laxmi Yeruva (vlyeruva@uams.edu)

Version: 3 Date: 19 Feb 2020

Author’s response to reviews:

Editor Comments:

1. Please provide/copy all relevant ethics information, including approval by the appropriate committee into the "ethics approval and consent to participate" section of your Declarations.

A: As recommended we have added the sentence to ethics approval and consent to participate section (lines 332-335).

2. Thank you for providing an Availability of data and materials statement for the Declarations section. Please describe where the raw data generated/analysed during your study can be found. Raw data does not include data that has been summarized in tables/figures. If you wish to share the data through correspondence then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.” If you do not wish to share your data, please state this along with the reason.

A: Thank you for the comment, we have added the text to share the data upon request (lines 339-340).
3. Please confirm in your response whether the animals obtained for your study were from a commercial source or somewhere else. Please confirm that the authors have gained approval from the provider of these animals to be used in the study.

A: In both methods and in ethics approval section the commercial source of the animals and approval of the farm text has been added (lines 96-97, 332-335).

4. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

A: No highlights have been added to the new text as per the recommendation.